Author's response to reviews

Title: Corticosteroid and antiviral therapy for Bell palsy: A network meta-analyses

Authors:

Pawin Numthavaj (pawinpawin@gmail.com)
Ammarin Thakkinstian (raatk@mahidol.ac.th)
Charunthai Dejthevaporn (racdt@mahidol.ac.th)
John Attia (John.Attia@newcastle.edu.au)

Version: 2 Date: 7 September 2010

Author’s response to reviews:

Dear Editor,

Thank you so much for giving us opportunity to revise the paper. We have revised it and covered all reviewer's comments. Please don’t hesitate to let us know if there is any further question.

Sincerely,
Ammarin